Author's response to reviews

Title: Prevalence of chronic diseases and morbidity in primary health care in central Greece: An epidemiological study

Authors:

Markos Minas (markosminas@hotmail.com)
Nikolaos Koukosias (nkouk@med.uth.gr)
Elias Zintzaras (zintza@med.uth.gr)
Konstantinos Kostikas (ktk@otenet.gr)
Konstantinos Gourgoulianis (kgourg@med.uth.gr)

Version: 4 Date: 18 August 2010

Author's response to reviews: see over
Dear Editor,

Thank you for the acceptance of our manuscript. According to the editorial production team suggestions about the format of our manuscript the following changes have been made:

Formatting changes to be made:
Major revisions (we require the author to make these changes)

Authors' contributions: Please indicate that all authors read and approved the final manuscript.
Response: The comment was added to the author’s contributions section

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Coloured text: please ensure that all text is black.
Response: All text is black

Materials and methods: please rename this section 'Methods'.
Response: The section was renamed “Methods”

Abbreviations: Please format these in to a paragraph.
Response: Abbreviations were formatted into a paragraph

Tables- please remove the visible vertical lines from your tables
Response: All vertical lines were removed

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
Response: The manuscript was checked for typographical errors
Moreover, we would like to inform you that Author's names and affiliations were checked and the affiliation of Ass. Prof. Elias Zintzaras was added.

We are at your disposal for any further clarifications.

Yours sincerely,

Dr Markos Minas, MD
Respiratory Medicine Department
University of Thessaly Medical School